\begin{table}%t6.a
\caption{\label{table:photo-scra}Photometric flux measurement of S~CrA~N and S~CrA~S. For the photometric observations of wavelengths $\lambda<0.64~{\rm \mu m}$ and $\lambda>12~{\rm \mu m}$, both components could not be spatially resolved.}
 %\centerline
 {\small   \begin{tabular}{lllr}
\hline\hline
      & S~CrA~N & S~CrA~S & \\
      Wavelength [${\rm \mu m}$] & Flux [Jy] & Flux [Jy] & Ref.\\
\hline
      $0.44$ &   \multicolumn{2}{c}{$0.0089$} & { 1} \\
      $0.55$ &   \multicolumn{2}{c}{$0.13$} & { 2} \\
      $0.64$ &   \multicolumn{2}{c}{$0.09$} & { 1} \\
      $1.25	$ & $0.57 \pm 0.02$ & $0.28 \pm 0.02$ & { 3} \\
      $1.65	$ & $1.02 \pm 0.04$ & $0.50 \pm 0.03 $ & { 3} \\
      $2.22	$ & $2.44 \pm 0.21$ & $0.74 \pm 0.09 $ & { 4} \\
      $3.50	$ & $3.99 \pm 0.23$ & $1.15 \pm 0.03 $ & { 4} \\
      $8.80	$ & $2.90 \pm 0.20$ & $1.59 \pm 0.17 $ & { 4}\\
      $10.6	$ & $3.19 \pm 0.20$ & $1.27 \pm 0.06$ & { 4} \\
      $12.0	$ & \multicolumn{2}{c}{$3.97$} & { 5} \\
      $12.0	$ & \multicolumn{2}{c}{$4.88 \pm 0.59$} & { 5} \\
      $12.5	$ & \multicolumn{2}{c}{$1.50 \pm 0.10$} & { 5} \\
      $25	$ & \multicolumn{2}{c}{$9.21 \pm 0.64$} & { 5} \\
      $25	$ & \multicolumn{2}{c}{$8.52 $} & { 5} \\
      $60	$ & \multicolumn{2}{c}{$1.81 \pm 0.16$} & { 5} \\          
      $60 $ & \multicolumn{2}{c}{$2.7 \pm 4.5 $} & { 6}\\   
      $60 $ & \multicolumn{2}{c}{$3.32 \pm 3.1$} & { 6} \\
      $90 $ & \multicolumn{2}{c}{$2.87 \pm 2.85$} & { 6} \\      
      $100$ & \multicolumn{2}{c}{$2.79 \pm 0.75$} & { 5} \\
      $170$ & \multicolumn{2}{c}{$8.29 \pm 22.24$} & { 6} \\      
      $450$ & \multicolumn{2}{c}{$2.75 \pm 0.38$} & { 7} \\
      $800$ & \multicolumn{2}{c}{$0.77 \pm 0.05$} & { 7} \\
      $1100$ & \multicolumn{2}{c}{$0.36 \pm 0.03$} & { 7} \\
      \hline
    \end{tabular}}
\medskip
References~-- {1}: Denis data base~(\cite{denis});  {2}: Takami et~al.~(\cite{takami}); {3}: Prato et~al. (\cite{prato}); {4}:~Mc~Cabe et~al. (\cite{cabe}); {5}:~IRAS catalogue (\cite{iras});  {6}:~ISO data archive; {7}: Jensen et~al.(\cite{jensenII}). 
         \end{table}